

Citation: Ledien J, Cucunubá ZM, Parra-Henao G, Rodríguez-Monguí E, Dobson AP, Adamo SB, et al. (2022) Linear and Machine Learning modelling for spatiotemporal disease predictions: Force-of-Infection of Chagas disease. PLoS Negl Trop Dis 16(7): e0010594. https://doi.org/10.1371/journal. pntd.0010594

Editor: Uwem Friday Ekpo, Federal University of Agriculture Abeokuta, NIGERIA

Received: February 11, 2022

Accepted: June 18, 2022

Published: July 19, 2022

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0010594

Copyright: © 2022 Ledien et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The datasets used for these analyses are available in the GitHub repository for Chagas disease Fol with Linear RESEARCH ARTICLE

Linear and Machine Learning modelling for spatiotemporal disease predictions: Force-of-Infection of Chagas disease

Julia Ledien^{1*}, Zulma M. Cucunubá^{2,3}, Gabriel Parra-Henao^{4,5}, Eliana Rodríguez-Monguí⁶, Andrew P. Dobson⁷, Susana B. Adamo⁸, María-Gloria Basáñez², Pierre Nouvellet¹

 School of Life Sciences, University of Sussex, Falmer, Brighton, United Kingdom, 2 London Centre for Neglected Tropical Disease Research & MRC Centre for Global Infectious Disease Analysis, School of Public Health, Imperial College London, London, United Kingdom, 3 Departamento de Epidemiología Clínica y Bioestadística, Facultad de Medicina, Universidad Pontificia Javeriana, Bogotá, Colombia, 4 Centro de Investigación en Salud para el Trópico, Universidad Cooperativa de Colombia, Santa Marta, Colombia, 5 National Institute of Health, Bogotá, Colombia, 6 Independent consultant to the Neglected, Tropical and Vector Borne Diseases Program, Pan American Health Organization (PAHO), Bogota, Colombia, 7 Department of Ecology and Evolutionary Biology, Princeton University, Princeton, New Jersey, United States of America, 8 Center for International Earth Science Information Network (CIESIN), Columbia Climate School, Columbia University, New York, New York, United States of America

* j.ledien@sussex.ac.uk

Abstract

Background

Chagas disease is a long-lasting disease with a prolonged asymptomatic period. Cumulative indices of infection such as prevalence do not shed light on the current epidemiological situation, as they integrate infection over long periods. Instead, metrics such as the Forceof-Infection (FoI) provide information about the rate at which susceptible people become infected and permit sharper inference about temporal changes in infection rates. FoI is estimated by fitting (catalytic) models to available age-stratified serological (ground-truth) data. Predictive FoI modelling frameworks are then used to understand spatial and temporal trends indicative of heterogeneity in transmission and changes effected by control interventions. Ideally, these frameworks should be able to propagate uncertainty and handle spatiotemporal issues.

Methodology/principal findings

We compare three methods in their ability to propagate uncertainty and provide reliable estimates of Fol for Chagas disease in Colombia as a case study: two Machine Learning (ML) methods (Boosted Regression Trees (BRT) and Random Forest (RF)), and a Linear Model (LM) framework that we had developed previously. Our analyses show consistent results between the three modelling methods under scrutiny. The predictors (explanatory variables) selected, as well as the location of the most uncertain Fol values, were coherent across frameworks. RF was faster than BRT and LM, and provided estimates with fewer extreme Models. Available from: https://github.com/jledien/ Chagas-disease-Fol-with-Linear-Models.git.

Funding: J.L. was funded by the award of a School of Life Sciences, University of Sussex PhD Studentship (to P.N.). M.G.B. acknowledges funding from the Medical Research Council (MRC) Centre for Global Infectious Disease Analysis (MR/ R015600/1), jointly funded by the UK MRC and the UK Foreign, Commonwealth & Development Office (FCDO), under the MRC/FCDO Concordat agreement and is also part of the European and Developing Countries Clinical Trials Partnership (EDCTP2) programme supported by the European Union. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

values when extrapolating to areas where no ground-truth data were available. However, BRT and RF were less efficient at propagating uncertainty.

Conclusions/significance

The choice of Fol predictive models will depend on the objectives of the analysis. ML methods will help characterise the mean behaviour of the estimates, while LM will provide insight into the uncertainty surrounding such estimates. Our approach can be extended to the modelling of Fol patterns in other Chagas disease-endemic countries and to other infectious diseases for which serosurveys are regularly conducted for surveillance.

Author summary

Metrics such as the per susceptible rate of infection acquisition (Force-of-Infection) are crucial to understand the current epidemiological situation and the impact of control interventions for long-lasting diseases in which the infection event might have occurred many years previously, such as Chagas disease. FoI values are estimated from serological age profiles, often obtained in a few locations. However, when using predictive models to estimate the FoI over time and space (including areas where serosurveys had not been conducted), methods able to handle and propagate uncertainty must be implemented; otherwise, overconfident predictions may be obtained. Although Machine Learning (ML) methods are powerful tools, they may not be able to entirely handle this challenge. Therefore, the use of ML must be considered in relation to the aims of the analyses. ML will be more relevant to characterise the central trends of the estimates while Linear Models will help identify areas where further serosurveys should be conducted to improve the reliability of the predictions. Our approaches can be used to generate FoI predictions in other Chagas disease-endemic countries as well as in other diseases for which serological surveillance data are collected.

Introduction

Chagas disease is a neglected tropical disease estimated to affect between 6 and 7 million persons worldwide. While only endemic in 21 countries in Latin America, the number of Chagas disease cases detected in Europe, North America, and the Far East has greatly increased, due to migration of infected populations [1]. Being able to identify how the cases are distributed in space and whether the control interventions implemented have been successful is critical to identifying how resources should be allocated to eliminate the disease as a public health problem in the 2021–2030 time horizon [2]. As a long-lasting and chronic disease, analyses based solely on the current prevalence of infection (typically measured as seroprevalence) has limited scope. Indeed, the prevalence recorded at a given time does not reflect the current epidemiological situation, as infection may have occurred in the past. The Force-of-Infection (FoI), i.e. the rate at which susceptible individuals become infected, is a modelling-derived metric that can be used to understand changes in incidence in space and time as a result of deliberate control interventions and/or secular changes, including environmental change [3]. However, the use of FoI raises its own challenges, particularly those regarding quantification and propagation of uncertainty when used as a response variable in predictive models. A catalytic model (fitted to age-structured seroprevalence data, often using Bayesian methods) has been used to obtain the FoI and thus, the FoI values for each serosurvey and each year are posterior distributions and not only single values [4]. When the derived FoI is used to fit predictive models, the mean or median values of FoI are predominantly used, neglecting the uncertainty surrounding the estimated values [5–7]. Furthermore, when a non-constant (e.g. a yearly-varying) FoI is assumed, each serosurvey analysed becomes a temporal series at a certain location, requiring specific and computationally-intensive methods to be included into predictive models [8]. Machine Learning methods could represent a faster and more flexible framework to implement such models.

Machine Learning (ML) methods are computational processes based on probabilities and algorithms that use prior knowledge to produce predictions. ML can handle non-linear and non-parametric models that are able to flout the linearity, normality (Gaussian distribution) and equal variance assumptions of statistical models [9]. Essentially, ML methods make no assumptions about the statistical distribution of the data [9].

These methods have previously been used in contexts in which those assumptions are challenged, such as spatial, temporal and spatiotemporal analyses of infectious diseases, e.g. mapping of human leptospirosis [10,11], severe fever with thrombocytopenia syndrome [12], lymphatic filariasis [13], or to identify individuals with a higher risk of HIV infection based on socio-behavioural-driven data [14].

Two types of ML models have been extensively used in the context of infectious disease epidemiology, namely, Boosted Regression Trees (BRT) and Random Forest (RF). Although they are not spatial approaches (as data locations and sampling patterns are ignored to produce estimates), they have shown potential in spatial modelling [15], in particular, when used with appropriate sampling strategies [16]. Specifically, BRT and RF have been used to study the spatial spread of numerous infectious diseases, including epidemics among swine farms in the USA [17], Ebola case-fatality ratio [18], risk factors for visceral leishmaniasis [19,20], African swine fever [21], scrub typhus [22], dengue incidence [23], and dengue FoI [5]. RF also proved its potential in modelling epidemics in a spatiotemporal framework, outperforming time series models [17].

This paper aims to compare the performance of two ML methods, namely, BRT and RF, with a Linear Model (LM) framework we have previously developed [8] in their ability to predict the FoI of Chagas disease across space and time. We use detailed data from Colombia as a case study and describe the advantages and disadvantages of using such Machine Learning methods compared to Linear Model frameworks, specifically focussing on their ability to handle uncertainty on the response variable.

Methods

1. Data sources

Current and past exposure to Chagas disease can be characterised by estimating the (timevarying) Force-of-Infection (FoI), i.e. temporal changes in the per susceptible rate of parasite acquisition [3,4]. Using results of 76 age-stratified serosurveys conducted at municipal level in Colombia between 1998 and 2014 (S1 Fig), yearly-varying FoI values were estimated, for each serosurvey, by fitting a catalytic model to age-stratified seropositivity data (see [4] for details). For each serosurvey, FoI estimates, for the period ranging from the birth of the oldest participants to the year when the serosurvey was conducted, were obtained using a Bayesian framework to fit the catalytic model to data, thus allowing for extraction of the full joint posterior distribution of the yearly FoI estimates. We refer to those municipalities where at least one serosurvey was conducted as municipalities 'in catchment areas', whereas those municipalities for which serosurveys were not conducted, not available, or not used in our analyses, are referred as 'out of catchment areas'. <u>S1 Fig</u> in the Supporting Information file depicts the geographical distribution of the available serosurveys ('ground-truth' data).

The predictors used in these analyses included demographic, entomological and climatic variables (recorded at the municipality level), contextual information about the serosurveys (location, year conducted and setting, i.e. urban, rural, mixed and indigenous (as defined in [4]), and information from public blood banks on the prevalence of Chagas disease and number of blood units tested (available at the departmental level). A full list and description of the predictors is available in <u>S1 Table</u> of the Supporting Information File.

2. Linear Model (LM) framework

The LM framework relied on a list of plausible linear combinations of predictors selected based on expert knowledge and to avoid correlation between predictors. They were then integrated into an ensemble model using model averaging with weights based on the performance indicators of each individual linear model. The 10 best models for each setting type (urban, rural and indigenous) were averaged and used to obtain FoI predictions. The LM framework has been fully described in [8].

3. Machine Learning (ML) framework

Both ML methods tested in this paper are based on decision trees. A decision tree is an intuitive process that builds an algorithm by generating a step-by-step tree, whereby the dataset is repeatedly split to make a decision at each node. The splitting relies on optimising a variablespecific threshold that best discriminates the data into two branches at each node. Sequentially, the entire dataset is divided by defining new variable-specific thresholds defining the nodes in the decision tree.

The size of the tree, its complexity (reflecting predictors' interactions), the number of observations in the terminal nodes and the criteria to stop the process are defined as model hyperparameters and form the basis of more complex designs.

3.1. Boosted Regression Trees (BRT)

Boosting Regression Trees (BRT) or Gradient Boosting Trees (GBT) are based on the building of a large number of small decision trees. The boosting aspect refers to fitting repeatedly very simple and basic classifiers, in which a different subset of the data is used for fitting at each iteration [9]. The Gradient technique is used to reduce the variance in the model; sequentially, each new tree added to the model is fitted to explain the remaining variance from the previous trees.

While BRT is considered a robust ML method, including its use for spatiotemporal analyses [19,20,22,24], it is known as having a tendency to overfit, unless a very large amount of data is available [25].

3.2. Random Forest (RF)

Random Forest (RF), first described by Breiman in 2001 [26], consists of a large collection of decision trees [9]. To grow an RF tree, random inputs and predictors are selected at each node [26], and this randomness is thought to reduce overfitting. RF is also considered a robust ML method that can handle outliers and noise while being faster than bagging- and boosting-based methods [26].

RF is not explicitly designed to explore spatial observations [15], and is known to produce suboptimal prediction when sampling is spatially biased and/or in the presence of strong spatial correlation [15]. However, spatiotemporal resampling strategies and variable selection processes have been developed to overcome this challenge [16,27].

4. Models' workflow

In order to assess the importance of integrating uncertainty on the response variable, we implemented two approaches The former relies on generating and assessing model predictions using the *median* estimates of the FoI for each serosurvey as an outcome (referred to as Med-FoI). The latter seeks to propagate the uncertainty linked to the catalytic model-derived 'observations' by accounting for the *full posterior distribution* of the FoI estimates (referred to as FullPostFoI).

With the MedFoI approach, models are fitted to the median FoI estimates and the performance indicator, the predictive R^2 , is based on central tendencies only. With the FullPostFoI approach, models are fitted on the full posterior distribution of FoI estimates and the performance indicator is based both on central tendency and on the amount of overlap between the 'observed' and predicted distribution of the outcome. This allowed us to quantify the ability of the models to match the uncertainty surrounding the FoI estimates (i.e. the outcome) inherited from the catalytic model (Fig 1). The percentage of overlap was obtained using the "overlap" function from the Overlapping R-package [28] and provided the proportion of the area of two kernel density estimations that overlap [29].

For the two approaches, we defined six different coefficients of determination (R^2) linked to the sampling strategy. An R^2 was estimated based directly on the data used to fit the models; a predictive R^2 was estimated based on a proportion of the dataset that was not used to fit the models, i.e. the cross-validation (CV) set. In addition, both urban- and rural-specific predictive R^2 were estimated based on the urban/rural data from the CV set. Finally, in the ML frameworks, a resample R^2 was estimated based on out-of-sample data for each resample iterations (see Fig 2 for a schematic representation of these approaches).



Fig 1. Graphical representation of the two modelling approaches used for each of the frameworks tested. The upper panel corresponds to MedFoI (fitted on median FoI); the lower panel depicts the FullPostFoI (fitted on full posterior FoI). The predictive R^2 values are calculated on cross-validation sets for both approaches (see Fig 2). In the upper panel, the performance indicator, *Ind*, is the R^2 , based on central tendency alone; in the lower panel, both central tendency and percentage of overlap enter into the calculation of the performance indicator, *Ind* (as the arithmetic mean between R^2 and percentage overlap). The percentage of overlap (% of overlap) represents the proportion of the 'observed' and predicted distributions that overlap.

https://doi.org/10.1371/journal.pntd.0010594.g001



Fig 2. Description of the modelling workflow for the Linear Models (LM) and the Machine Learning (ML) frameworks. ML framework include Boosted Regression Trees (BRT) and Random Forest (RF) methods). CV denotes cross-validation; Pred R^2 urban and Pred R^2 rural denote urban- and rural-specific predictive R^2 values that were estimated based on the urban/rural data from the CV set; %Overlap indicates the proportion of the 'observed' and predicted distributions that overlap (see Fig 1), assessed over all settings and for urban and rural settings separately.

While the LM framework necessitated transformation of the data to normalise them, ML methods should, in principle, to be able to handle data without requiring normalisation (i.e. without requiring that their distribution is Gaussian). However, this process can help improve the performance of the model and was, therefore, tested (i.e. ML approaches were used to predict the FoI values both on non-transformed and log-transformed scales).

While the LM framework relies on a list of plausible and pre-defined linear models including interactions between factors (predictors) and excluding correlated predictors, the ML framework is implemented in two steps, to be fitted only on the ten most important variables. At first, ML models were fitted using all the predictors available, then the importance/influence of each predictor was assessed, and the 10 most influential factors were used in the second step, during which the models were fitted again, and predictions extracted.

Finally, ML requires a tuning step, during which the best hyperparameters are selected. Also, the resampling strategy used within the LM framework relies on random resampling while a spatiotemporal resampling strategy has been used for the ML framework. A detailed description of the tuning of hyperparameters and the comparison of several resampling strategies is available in <u>S1 Appendix</u> of the Supporting Information File, including details about the tuning of hyperparameters and the comparison of several resampling strate-

5. Indicators used to compare LM with ML frameworks

The best ML models obtained were then compared with the LM framework previously developed [8] in terms of their performance indicators, predictions, and ability to propagate uncertainty. In addition to these aspects, the models' ability to deal with temporal and spatial correlation, as well as their different computational aspects entered the comparison.

To allow comparison of predictive ability across multiple serosurveys, the distributions of predictions were standardised to the 'observations', allowing us to visualise whether the median and confidence intervals of the predictions matched those (median and credible intervals) of the catalytic model-derived FoI 'observations'. This process was performed at the sero-survey level to assess how much of the uncertainty inherited from the FoI calculation (via catalytic model fitting) was propagated into the predictions (see <u>S2 Appendix</u> in the Supporting Information File for details).

The uncertainty in the predictions was quantified using the Coefficient of Variation based on the standardised Median Absolute Deviation (MAD-CV), as FoI values were not normally distributed [30]. (Note that MAD-CV refers to coefficient of variation, whilst CV denotes cross validation.)

The residual spatial correlation was assessed using the Moran's *I* heterogeneity test from the "spdep" R package [31]. For the LM framework, the Moran's *I* test was applied on all the residuals (originating from the cross validation (CV) and fitting sets) excluding those from the rural–urban mixed settings (as LM model selection was based on setting type and no model was explored, selected or averaged for mixed settings, and thus no predictions were produced for the 'observed' FoI values corresponding to such settings). For ML models, the Moran's *I* test was applied to the residuals of the CV set. Residuals for a single year were used to exclude potential temporal autocorrelation, and for presentation purposes, we selected 2005 as the year with the largest number of independent FoI 'observations'.

The residual temporal correlation was tested using a Durbin-Watson test (DW) [32] (see Eq 1 for the DW statistic). In order to capture the residual correlation inherited from the estimation of the FoI values through fitting the catalytic model, the residuals being compared were always from the same serosurvey and for consecutive years. Thus, the DW statistic provided the residual serial correlation for a lag of one year,

$$DW = \frac{\sum_{i=0}^{n} (r_i - lag(r_i))^2}{\sum_{i=0}^{n} r_i^2}$$
Eq1

where *r* denotes the residuals for *i* serosurveys, *lag* is one year (for consecutive, yearly series of serosurveys), and *n* the number of 'observations' tested.

| | | BRT | | | | RF | | | | | | |
|--------------------|--------------|-----|-------|-----|-------|-----|--------------|-----|-------|-----|-------|-----|
| | All settings | | Urban | | Rural | | All settings | | Urban | | Rural | |
| | non | log | non | log | non | log | non | log | non | log | non | log |
| MedFoI | | | | | | | | | | | | |
| R ² (%) | 98 | 95 | 95 | 96 | 94 | 90 | 98 | 96 | 90 | 98 | 93 | 93 |
| Overlap | 23 | 19 | 21 | 19 | 25 | 19 | 22 | 21 | 20 | 21 | 25 | 21 |
| FullPostFoI | | | | | | | | | | | | |
| R^{2} (%) | 60 | 58 | 53 | 68 | 39 | 68 | 63 | 59 | 63 | 68 | 69 | 70 |
| Overlap | 25 | 36 | 24 | 34 | 22 | 36 | 40 | 42 | 42 | 42 | 40 | 42 |
| Indicator | 43 | 48 | 37 | 52 | 25 | 52 | 51 | 50 | 50 | 55 | 52 | 55 |

Table 1. Median cross-validation performance values for the two Machine Learning modelling methods investigated.

MedFoI: models fitted on median FoI; *FullPostFoI*: models fitted on full posterior FoI, without (non) or with log-transformation (log) of Force-of-infection (FoI) 'observations' (generated by fitting catalytic models to age-stratified serological surveys for Chagas disease in Colombia, with yearly-varying FoI [8]).

BRT: Boosted Regression Trees; RF: Random Forest methods; performance indicators are reported for either all settings (urban, rural, indigenous and mixed), urban, or rural settings separately. The predictive R^2 values were calculated on cross-validation datasets and are expressed as percentage.

Overlap: proportion (expressed as percentage) of 'observed' and predicted distributions that overlaps (reflective of the degree of dispersion around central tendency). For *MedFoI* models, the performance indicator is the value of R^2 alone. Therefore % overlap is presented for comparison but was not used in fitting or selecting models. For *FullPostFoI*, the performance indicator is the arithmetic mean between R^2 and % of overlap (see Fig 1).

https://doi.org/10.1371/journal.pntd.0010594.t001

All ML analyses were run under the mlr3 framework (an object-oriented machine learning framework in R) [33] using R-4.0.3 software [34].

Results

1. Comparison of the performance of LM and ML frameworks

The predictive R^2 values for the LM framework obtained, on average, for its 5 best-fitting models, were 77% and 70%, with %overlap of 54% and 39% for urban and rural settings, respectively [8]. For the ML frameworks, the MedFoI approach yielded substantially better predictive R^2 values (ranging between 90% and 98%), but the degree of overlap between the distributions of the FoI 'observations' and the predictions was substantially lower (19%–25%), reflective of a tighter distribution around the central estimates and thus indicating over-confidence in the predictions when using such a simple approach (i.e. an approach that ignores the uncertainty linked to the outcome). The FullPostFoI approach gave a more balanced performance indicator, with predictive R^2 values ranging between 39% and 70%, and %overlap between 22% and 42% (Table 1). For both BRT and RF methods, the use of log-transformation to normalise the distribution of the FoI 'observations' consistently led to improved results (Table 1), with predictive R^2 values ranging between 59% and 70%, and %overlap between 34% and 42%.

Nested resampling, tested on the RF method with log-transformation, did not substantially improve model performance. Thus, the following subsections focus on the results obtained by fitting the frameworks on the full posterior distribution of the log-transformed FoI.

2. Comparison of the influence of predictor variables

The factors selected for the ML models were consistent with those that had been selected for the LM framework (Table 2); a Spearman correlation test showed that there was substantial rank correlation of the predictors included among the three models investigated (with Spearman rank correlation coefficient, r_s , between LM and BRT = 0.50; between LM and RF = 0.54, and between BRT and RF = 0.64, all p-values <0.05).

| Table 2. Standardized relative influence, importance and rank of the predictors included in Boosted Regression Trees (BRT) and Random Forest (RF) Machine |
|---|
| Learning models and normalised number of times the predictors were used in the linear model (LM) framework and their rank when using the full posterior distri- |
| bution of FoI estimates. |

| | Predictors | BRT | | RF | | LM | |
|---------|---|-----------|------|------------|------|------|------|
| Code | Name | Influence | Rank | Importance | Rank | Used | Rank |
| | Serosurvey characteristics: | | | _ | | | |
| S01 | Year of the survey | 0.20 | 2 | 0.17 | 1 | 0.05 | 4 |
| S02 | Rural setting | 0.03 | 11 | 0.03 | 11 | 0.14 | 2 |
| S03 | Urban setting | 0.03 | 10 | 0.04 | 7 | 0.15 | 1 |
| S04 | Indigenous setting | 0.20 | 1 | 0.12 | 4 | 0.15 | 1 |
| S05 | Latitude | 0.14 | 3 | 0.16 | 2 | 0.14 | 2 |
| S06 | Longitude | 0.04 | 8 | 0.09 | 5 | 0.02 | 7 |
| | Blood-bank data: | - | | | | | |
| B01 | Seroprevalence | 0.00 | NU | 0.03 | 14 | 0.04 | 5 |
| B02 | Proportion of blood units screened | 0.00 | NU | 0.03 | 10 | 0.01 | 8 |
| | Demography: | | | | | | |
| D01 | Population density | 0.10 | 4 | 0.13 | 3 | 0.01 | 8 |
| D02 | Poverty | 0.07 | 6 | 0.01 | 15 | 0.04 | 5 |
| D03 | Rural Indigenous Population size | 0.00 | NU | 0.03 | 9 | 0.00 | NU |
| | Climate: | | | | | | |
| | Continuous | 1 | | | | | |
| C01 | Polar climate frequency | 0.03 | 12 | 0.00 | 18 | 0.04 | 5 |
| C02 | Tropical climate frequency | 0.03 | 9 | 0.04 | 8 | 0.01 | 8 |
| C03 | Temperate climate frequency | 0.04 | 7 | 0.00 | 17 | 0.00 | NU |
| C04 | Arid climate frequency | 0.00 | NU | 0.00 | 21 | 0.00 | NU |
| | Categorical | | | | | | |
| C05 | Tropical climate categorised | NT | | | | 0.06 | 3 |
| C06 | Polar Climate Presence | 0.00 | NU | 0.00 | NU | 0.00 | NU |
| | Entomological data: | | | | | | |
| | At departmental level | | 1 | | | | |
| V01 | R. prolixus geographical extent | 0.00 | NU | 0.06 | 6 | 0.02 | 7 |
| V02 | <i>T. dimidiata</i> geographical extent | 0.00 | NU | 0.03 | 12 | 0.01 | 8 |
| V03 | R. prolixus presence | 0.00 | NU | 0.00 | NU | 0.00 | NU |
| V04 | T. dimidiata presence | 0.00 | NU | 0.00 | NU | 0.03 | 6 |
| | At municipality level | | | | | | |
| V05 | R. prolixus density | 0.00 | 13 | 0.00 | NU | 0.01 | 8 |
| V06 | T. dimidiata density | 0.00 | 15 | 0.00 | 16 | 0.01 | 8 |
| V07 | R. prolixus presence | 0.00 | NU | 0.00 | NU | 0.00 | NU |
| V08 | <i>T. dimidiata</i> presence | 0.00 | NU | 0.00 | 20 | 0.00 | NU |
| | Interventions: | | | | | | |
| | At municipality level | | 1 | | | | |
| I01 | Intervention intensity | 0.00 | 14 | 0.00 | NU | 0.00 | NU |
| 102 | Intervention category | NT | | NT | | 0.01 | 8 |
| | At household level | | | | | | |
| I03 | Household intervention | 0.00 | NU | 0.00 | NU | 0.00 | NU |
| I04 | Household intervention category | NT | | NT | | 0.01 | 8 |
| | Temporal factors: | | | | | | |
| T01 | Year | 0.09 | 5 | 0.03 | 13 | 0.01 | 8 |
| - • • | | 5.07 | | 0.00 | 10 | 0.01 | |

(Continued)

Table 2. (Continued)

| Predictors | | BRT | | RF | | LM | |
|------------|--------|-----------|------|------------|------|------|------|
| Code | Name | Influence | Rank | Importance | Rank | Used | Rank |
| T02 | Decade | 0.00 | NU | 0.00 | NU | 0.00 | NU |

NU: Not used in the model; NT: not tested in the model.

R. prolixus: Rhodnius prolixus; T. dimidiata: Triatoma dimidiata.

The shade of green is associated to the rank of the predictors with darker green predictors having more importance.

https://doi.org/10.1371/journal.pntd.0010594.t002

The type of setting was the most important factor for both the LM and BRT. Latitude, year when the serosurvey was conducted, and population density also played an important role. Poverty, climatic and entomological features had a moderate role.

For the ML frameworks, blood-bank and intervention-related features were less influential than for the LM framework. Generally, the year (temporal trend) seemed to play a greater role in the ML models.

3. Comparison of spatial trends in predictions

All methods showed generally similar spatial trends and comparable levels of uncertainty (Fig 3) for FoI prediction across Colombia (using the year 1990 for illustration as the pattern is consistent in time). Generally, FoI estimates were higher in northern and eastern municipalities and lower in the south of the country, with the latter being associated with higher uncertainty (Fig 3). The BRT framework predictions showed increased spatial heterogeneity, while predictions from the LM framework resulted in more spatially uniform predictions.

When comparing FoI predictions directly across the three methods, for urban and rural settings (Fig 4), we found good agreement between all of them, particularly between RF and LM. Generally, the BRT tended to predict higher FoI values in both settings. The patterns observed in the entire country seemed to follow what was observed in the catchment areas (municipalities where at least one serosurvey was conducted).

4. Comparison of temporal trends in predictions across serosurveys

When comparing 'observations' with predictions over time, all methods performed well regarding their ability to capture central trends (Fig 5). However, the LM framework was better at capturing uncertainty, as the confidence bounds of the predictions mirrored more closely the credible intervals (CrI) of the 'observations'.

The median uncertainty across municipalities (Table 3) was comparable using any of the methods and restricting the assessment to 'in catchment area' only (i.e. municipalities where at least one serosurvey had been conducted) or not ('out of catchment area'). However, for some municipalities, the uncertainty associated with the LM framework increased dramatically.

Comparatively, the RF method produced more uniform uncertainty across predictions, with median and range similar to those yielded by the other (BRT and LM) methods, but with fewer municipalities with substantial uncertainty (defined as MAD-CV>2), and only a moderate number of municipalities with extreme uncertainty (defined as MAD-CV>5).

5. Residual spatial and temporal correlation

While the ML-based methods did not show any significant spatial correlation in their residuals, this was not the case with the LM framework (Table 4). For all models, the DW test's statistic (see Eq 1) showed a significant residual temporal correlation between residuals from the



Fig 3. Spatial distribution of the Force-of-Infection of Chagas Disease (per year and per susceptible individual), in Colombia. The predicted distribution was generated using two Machine Learning (Boosted Regression Trees (BRT) and Random Forest (RF)) methods and a Linear Model (LM) framework (main maps); the associated uncertainty (small map insets) presents the Median Absolute Deviation (MAD) Coefficient of Variation (MAD-CV). Predictions were obtained at the municipality level for urban and rural settings, in 1990. (Borders shapefile at the ADM2 level obtained from GADM, https://gadm.org/download_country.html).

same serosurvey, with a stronger effect for serosurveys conducted in indigenous settings (S2 Fig in Supporting Information File).

6. Computational aspects

Computationally, RF and BRT required the least effort (31 and 42 hrs respectively, on standard laptop, with an i7-8565U processor and 16.0 GB RAM) (Table 5). By contrast, although



Fig 4. Comparison of predicted Chagas disease Force-of-Infection (FoI) values for urban or rural settings at municipality level, in Colombia for the year 1990. The values were obtained by two Machine Learning (Boosted Regression Trees (BRT) and Random Forest (RF)) methods and a Linear Model (LM) framework using log-transformed FoI estimates from the FullPostFoI approach (see Models' workflow subsection in Methods and Fig 1 for a description of this approach). The upper panel presents the results for urban settings; the lower panel presents the results for rural settings. Purple-coloured dots denote municipalities where at least one serosurvey had been conducted ('in catchment area'); teal-coloured dots denote municipalities where no serosurveys had been conducted or were not included in our analyses ('outside catchment area'). The black solid diagonal line represents perfect agreement between the two frameworks being compared.

implementation of LM required far fewer R-packages than the ML framework, it took over twice the time to run when compared to RF (72 hr). Also, the computer hard-drive space that was required to store 'objects' and model outputs was about 20 times higher for LM than for



Fig 5. Standardised comparisons of 'observed' and predicted distributions across serosurveys and by setting type. Comparisons were made for two Machine Learning (Boosted Regression Trees (BRT) and Random Forest (RF)) methods (upper and middle panels) and a Linear Model (LM) framework (lower panel) using log-transformed (log) Force-of-Infection (FoI) estimates from the FullPostFoI approach for urban and rural Chagas disease settings in Colombia across 9 decades. The solid lines and envelopes show standardised distances between FoI 'observations' and predictions, with purple-colour lines representing the median, and the pink and blue lines representing, respectively, the upper and lower bounds of the 95%CrI. If medians and 95% confidence bounds of the predictions matched exactly the corresponding measures for all the 'observations' across serosurveys, then the solid and dashed lines would fully overlap.

the ML framework. Finally, the overall implementation of the models was substantially simpler for the ML framework; particularly to make adjustments and updates.

Discussion

Our comparative analyses indicated generally consistent results among the three modelling methods investigated to generate Chagas disease FoI predictions, namely, the linear model (LM) framework we previously developed [8], and the two Machine Learning (ML) methods explored here (Boosted Regression Trees (BRT) and Random Forest (RF)). The predictors that were selected, as well as the location of the most uncertain FoI values were coherent and generally consistent among the three methods (Table 2 and Figs 3 and 4). Not entirely surprising, RF was faster to run than BRT and LM [26] (Table 5).

Table 3. Uncertainty across Chagas disease Force-of-Infection predictions for the three frameworks under comparison. The uncertainty was estimated using the Median Absolute Deviation Coefficient of Variation (MAD-CV) of the predictions for Colombia in 1990, in (urban and rural) areas where at least one serosurvey had been conducted ('in catchment area') and where no data were available or used in the analyses ('out of catchment area'). The number of municipalities where MAD-CV is greater than 2 (substantial uncertainty) or greater than 5 (extreme uncertainty) is also included.

| | | MAD-CV values (range) | | | | | Number of municipalities | | |
|-----|-------------------|-----------------------|-----------------------|-------------------|-------|----------|--------------------------|----------|--|
| | In catchment area | | Out of catchment area | | MAD-0 | MAD-CV>2 | | MAD-CV>5 | |
| | Urban | Rural | Urban | Rural | Urban | Rural | Urban | Rural | |
| BRT | 1.45 (0.31-7.16) | 1.54 (0.39-5.40) | 1.48 (0.31-7.41) | 1.48 (0.17-6.32) | 338 | 335 | 25 | 24 | |
| RF | 1.47 (0.48-5.28) | 1.45 (0.40-5.29) | 1.48 (0.47-5.24) | 1.49 (0.44-5.22) | 145 | 198 | 10 | 8 | |
| LM | 1.60 (0.70-2.73) | 1.29 (0.44-2.76) | 1.48 (0.32-8.19) | 1.50 (0.24–11.00) | 284 | 266 | 6 | 11 | |

BRT: Boosted Regression Trees; RF: Random Forest; LM: Linear Model.

https://doi.org/10.1371/journal.pntd.0010594.t003

Based on the performance indicators used, RF performed best (Table 1) but did less well when considering the propagation of uncertainty in the FoI inherited from the catalytic model (Fig 5). Also, RF generated fewer municipality-level predicted values with substantial or extreme uncertainty (Table 3). All methods, when fitted on the median FoI alone (MedFoI approach), were unable to capture the uncertainty in the response variable (the FoI 'observations' generated by fitting the catalytic model to the age-stratified serosurveys), leading to overconfident predictions (with high predictive R^2 values but smaller % of overlap values). This highlights an important issue not fully addressed in the literature, as most publications using FoI data to infer spatiotemporal patterns of infectious disease incidence tend to use the central FoI estimates alone to fit predictive models (i.e. using what we labelled here as the MedFoI approach). We argue that neglecting to appreciate and propagate the uncertainty inherent in their estimation [5–7] may therefore lead to significant over-confidence in predictions. This issue, already highlighted in our previous LM work [8], is not mitigated by implementing ML frameworks, and deserves careful consideration, not only from a methodological perspective, but importantly, when the results are applied in policy-relevant contexts [35].

Indeed, quantifying and communicating uncertainty in FoI appropriately is critical when the results of predictive models are used to inform stakeholders and public health programme managers on the level of certitude associated with exposure risk or number of cases. Thus areas/populations for which exposure has been certainly high or low can be differentiated from those with exposure levels or number of cases that necessitate further investigation due to highly uncertain estimation.

Even when the three methods showed good performance and generally good agreement at the serosurvey level (Fig 4), the residuals remained correlated in time (Table 4). Thus, the

Table 4. Spatial and temporal correlation test statistics and statistical significance of the spatial correlation test applied to the cross-validation residuals for the two Machine Learning (BRT, RF) and the Linear Model (LM) methods under consideration.

| | BRT | RF | LM† | |
|----------------------------|-------|-------|-------|--|
| Spatial correlation test: | | | | |
| Moran's I statistic | 0.00 | 0.00 | 0.06* | |
| Temporal correlation test: | | | | |
| DW statistic | 0.06* | 0.04* | 0.00* | |

BRT: Boosted Regression Trees; RF: Random Forest; LM: Linear Model.

DW: Durbin-Watson statistic (see Eq 1).

†See methods for calculation of the LM residual correlation.

*p-values significant et 5%.

https://doi.org/10.1371/journal.pntd.0010594.t004

Table 5. Comparison of computational aspects for the Machine Learning (Boosted Regression Trees (BRT), Random Forest (RF)) and Linear Model (LM) methods investigated. The methods under comparison used log-transformed FoI values from the FullPostFoI approach.

| | BRT | RF | LM |
|--------------------------------------|------|------|-------|
| Number of R packages needed | 20 | 20 | 6 |
| Time required for models to run (hr) | 42.5 | 31.0 | 72.0 |
| Hard-drive space requirements (MB) | 149 | 114 | 2,048 |

BRT: Boosted Regression Trees; RF: Random Forest; LM: Linear Model. hr: hours; MB: Megabytes.

https://doi.org/10.1371/journal.pntd.0010594.t005

correlation inherited from the FoI calculation was not fully accounted for in any of our methods, i.e. none of the predictors included was able to account for the full extent of this correlation.

While the final ML models showed no evidence of residual spatial correlation (Table 4), the spatial extrapolation shown (Fig 3) should be interpreted with caution, as the (ground-truth) serosurveys available had only been conducted in a relatively small number of municipalities and tended to be aggregated in the same area (S1 Fig). When using RF, the degree of uncertainty inside and outside 'catchment areas' was consistent, suggesting reliable extrapolation. This contrasted with the LM framework, which predicted large uncertainty in some municipalities.

Most of the earliest serosurveys (up to early 2000) seemed to have targeted high-risk populations [36], presumably because the perceived risk of Chagas disease transmission in those areas was higher and required improved situational awareness. However, using only such information to make predictions across Colombia would have led to higher predicted FoI in areas where no ground-truth data had been collected. By contrast, the most recent serosurveys (2010–2014) seem to have been conducted on more representative samples of the population, presumably motivated by providing a more realistic assessment of the epidemiological situation nationally and demonstrating progress in reducing vectorial transmission. We, therefore, included the year when the serosurvey took place to account for this bias and, in fact, this variable appeared to be one of the most influential in all three methods and particularly for BRT and RF (Table 2). This demonstrates the crucial importance of understanding the motivation behind the implementation of serosurveys in order to assess the sampling strategy and ultimately quantify potential biases that may interfere with the representativeness of FoI estimates.

Finally, and regarding computational aspects, the LM framework required substantial userinput to prepare the data for model fitting (including data transformation; choice of predictors included in each model; tests for spatial and temporal correlation, etc.). In contrast, ML frameworks were faster (particularly RF) and required less pre-processing of the data and hard-drive space (Table 5). These features render the ML models more flexible, more readily updatable, and thus easier and simpler to be extended to other Chagas disease-endemic countries, and potentially to other infectious diseases, including neglected tropical diseases, for which serological surveys are regularly conducted as surveillance tools to assess epidemiological situation, incidence, and impact of control interventions across spatial and temporal scales [5,6,37,38].

Concluding remarks

ML methods are increasingly used to derive computationally efficient algorithms for data analysis that are agnostic to the distributional properties of such data. They represent an attractive modelling tool for the generation of predictive maps of important infectious disease epidemiological metrics, such as the FoI. Most published literature on the subject use measures of FoI central tendency, neglecting to quantify, propagate and ultimately communicate the uncertainty appropriately. We show that the uncertainty on the input variables cannot be neglected whatever statistical method is used and furthermore that the choice of modelling framework requires careful consideration according to the ultimate objectives of the modelling endeavour. If the aim is, for instance, to use the predicted FoI patterns to provide numbers of cases and estimates of the associated disease burden, ML framework (and particularly RF) would indeed be an optimal choice, as capturing the median (central tendency behaviour) may be sufficient and computationally efficient. However, if the objective is to identify areas where serological surveillance surveys are scarce and should be conducted to improve the reliability of FoI estimates and provide ground-truth data, we conclude that the LM framework, albeit more timeconsuming and computationally intensive, would provide a better indication of where uncertainty is greatest. Although in this paper we focused on Chagas disease in Colombia as a case study, the modelling frameworks compared here can be applied to other Chagas diseaseendemic countries and to infectious diseases (including neglected tropical diseases) for which age-stratified serological data are regularly collected.

Supporting information

S1 Appendix. Machine Learning tuning. (DOCX)

S2 Appendix. Comparing observations and predictions across serosurveys. (DOCX)

S1 Table. Variables tested as factors in the geospatial analyses of Chagas disease in Colombia.

(DOCX)

S1 Fig. Locations and sample sizes of Chagas disease serosurveys conducted in Colombia with information on the location at the municipality level, from 1998 to 2014 (Borders shapefile at the ADM2 level obtained from GADM, https://gadm.org/download_country. html). Figure extracted from (2). The grey boundaries delimitate the Departments. (TIF)

S2 Fig. Comparison of residual serial correlation for Boosted Regression Trees (BRT), Random Forest (RF) and Linear Model (LM) models built on the log scale based on the FullPostFoI approach. Each line corresponds to one serosurvey. (TIF)

Acknowledgments

The authors thank the Pan American Health Organization (PAHO) and in particular Dr Luis Gerardo Castellanos for their support. Our thanks also go to Dr Alpha Forna for valuable discussions.

Author Contributions

Conceptualization: Julia Ledien, Zulma M. Cucunubá, Gabriel Parra-Henao, Eliana Rodríguez-Monguí, Andrew P. Dobson, Susana B. Adamo, María-Gloria Basáñez, Pierre Nouvellet.

Data curation: Julia Ledien, Zulma M. Cucunubá, Gabriel Parra-Henao, Eliana Rodríguez-Monguí, Pierre Nouvellet.

Formal analysis: Julia Ledien.

Funding acquisition: Zulma M. Cucunubá, Pierre Nouvellet.

- Investigation: Julia Ledien, Zulma M. Cucunubá, Gabriel Parra-Henao, Eliana Rodríguez-Monguí, Andrew P. Dobson, Susana B. Adamo, María-Gloria Basáñez, Pierre Nouvellet.
- Methodology: Julia Ledien, Zulma M. Cucunubá, Gabriel Parra-Henao, Eliana Rodríguez-Monguí, Andrew P. Dobson, Susana B. Adamo, María-Gloria Basáñez, Pierre Nouvellet.
- Supervision: Zulma M. Cucunubá, María-Gloria Basáñez, Pierre Nouvellet.

Validation: Julia Ledien.

Visualization: Julia Ledien.

Writing - original draft: Julia Ledien.

Writing – review & editing: Julia Ledien, Andrew P. Dobson, María-Gloria Basáñez, Pierre Nouvellet.

References

- 1. WHO. Chagas disease [Internet]. [cited 2021 Dec 14]. Available from: https://www.who.int/news-room/ fact-sheets/detail/chagas-disease-(american-trypanosomiasis)
- World Health Organization. Ending the neglect to attain the sustainable development goals: a road map for neglected tropical diseases 2021–2030 [Internet]. Geneva: World Health Organization; 2020 [cited 2022 Feb 1]. 177 p. Available from: https://apps.who.int/iris/handle/10665/338565
- Muench H. Catalytic Models in Epidemiology [Internet]. Harvard University Press; 2013. Available from: https://doi.org/10.4159/harvard.9780674428928
- Cucunubá ZM, Nouvellet P, Conteh L, Vera MJ, Angulo VM, Dib JC, et al. Modelling historical changes in the force-of-infection of Chagas disease to inform control and elimination programmes: application in Colombia. BMJ Glob Health. 2017; 2(3):e000345. https://doi.org/10.1136/bmjgh-2017-000345 PMID: 29147578
- Cattarino L, Rodriguez-Barraquer I, Imai N, Cummings DAT, Ferguson NM. Mapping global variation in dengue transmission intensity. Sci Transl Med. 2020 Jan 29; 12(528):eaax4144. https://doi.org/10. 1126/scitranslmed.aax4144 PMID: 31996463
- Garske T, Van Kerkhove MD, Yactayo S, Ronveaux O, Lewis RF, Staples JE, et al. Yellow Fever in Africa: Estimating the burden of disease and impact of mass vaccination from outbreak and serological data. Hay SI, editor. PLoS Med. 2014 May 6; 11(5):e1001638. <u>https://doi.org/10.1371/journal.pmed.</u> 1001638 PMID: 24800812
- O'Driscoll M, Imai N, Ferguson NM, Hadinegoro SR, Satari HI, Tam CC, et al. Spatiotemporal variability in dengue transmission intensity in Jakarta, Indonesia. PLoS Negl Trop Dis. 2020 Mar 6; 14(3): e0008102. https://doi.org/10.1371/journal.pntd.0008102 PMID: 32142516
- Ledien J, Cucunubá ZM, Parra-Henao G, Rodríguez-Monguí E, Dobson AP, Basáñez MG, et al. Spatiotemporal variations in exposure: Chagas disease in Colombia as a case study. BMC Med Res Methodol. 2022 Jan 13; 22(1):13. https://doi.org/10.1186/s12874-021-01477-6 PMID: 35027002
- 9. Malley JD, Malley KG, Pajevic S. Statistical learning for biomedical data. Cambridge: Cambridge University Press; 2011. 285 p. [Practical guides to biostatistics and epidemiology].
- Ledien J, Sorn S, Hem S, Huy R, Buchy P, Tarantola A, et al. Assessing the performance of remotelysensed flooding indicators and their potential contribution to early warning for leptospirosis in Cambodia. PLOS ONE. 2017 Jul 13; 12(7):e0181044. https://doi.org/10.1371/journal.pone.0181044 PMID: 28704461
- Mohammadinia A, Saeidian B, Pradhan B, Ghaemi Z. Prediction mapping of human leptospirosis using ANN, GWR, SVM and GLM approaches. BMC Infect Dis [Internet]. 2019 Nov 13 [cited 2020 Feb 12]; 19. Available from: https://www.ncbi.nlm.nih.gov/pmc/articles/PMC6854714/ https://doi.org/10.1186/ s12879-019-4580-4 PMID: 31722676

- Miao D, Dai K, Zhao GP, Li XL, Shi WQ, Zhang JS, et al. Mapping the global potential transmission hotspots for severe fever with thrombocytopenia syndrome by machine learning methods. Emerg Microbes Infect. 2020 Dec; 9(1):817–26. https://doi.org/10.1080/22221751.2020.1748521 PMID: 32212956
- Eneanya OA, Fronterre C, Anagbogu I, Okoronkwo C, Garske T, Cano J, et al. Mapping the baseline prevalence of lymphatic filariasis across Nigeria. Parasit Vectors. 2019 Sep 16; 12(1):440. https://doi. org/10.1186/s13071-019-3682-6 PMID: 31522689
- Mutai CK, McSharry PE, Ngaruye I, Musabanganji E. Use of machine learning techniques to identify HIV predictors for screening in sub-Saharan Africa. BMC Med Res Methodol. 2021 Jul 31; 21(1):159. https://doi.org/10.1186/s12874-021-01346-2 PMID: 34332540
- Hengl T, Nussbaum M, Wright MN, Heuvelink GBM, Gräler B. Random forest as a generic framework for predictive modeling of spatial and spatio-temporal variables. PeerJ. 2018 Aug 29; 6:e5518. https:// doi.org/10.7717/peerj.5518 PMID: 30186691
- Meyer H, Reudenbach C, Wöllauer S, Nauss T. Importance of spatial predictor variable selection in machine learning applications–Moving from data reproduction to spatial prediction. Ecol Model. 2019 Nov 1; 411:108815.
- Augusta C, Deardon R, Taylor G. Deep learning for supervised classification of spatial epidemics. Spat Spatio-Temporal Epidemiol [Internet]. 2018 Aug 29; Available from: http://www.sciencedirect.com/ science/article/pii/S1877584517301636
- Forna A, Nouvellet P, Dorigatti I, Donnelly CA. Case Fatality Ratio Estimates for the 2013–2016 West African Ebola Epidemic: Application of Boosted Regression Trees for Imputation. Clin Infect Dis. 2020 Jun 10; 70(12):2476–83. https://doi.org/10.1093/cid/ciz678 PMID: 31328221
- Jiang D, Ma T, Hao M, Qian Y, Chen S, Meng Z, et al. Spatiotemporal patterns and spatial risk factors for visceral leishmaniasis from 2007 to 2017 in Western and Central China: A modelling analysis. Sci Total Environ. 2021 Apr 10; 764:144275. <u>https://doi.org/10.1016/j.scitotenv.2020.144275</u> PMID: 33385656
- Ding F, Wang Q, Fu J, Chen S, Hao M, Ma T, et al. Risk factors and predicted distribution of visceral leishmaniasis in the Xinjiang Uygur Autonomous Region, China, 2005–2015. Parasit Vectors. 2019 Nov 8; 12(1):528. https://doi.org/10.1186/s13071-019-3778-z PMID: 31703720
- Andraud M, Bougeard S, Chesnoiu T, Rose N. Spatiotemporal clustering and Random Forest models to identify risk factors of African swine fever outbreak in Romania in 2018–2019. Sci Rep. 2021 Jan 22; 11(1):2098. https://doi.org/10.1038/s41598-021-81329-x PMID: 33483559
- 22. Yao H, Wang Y, Mi X, Sun Y, Liu K, Li X, et al. The scrub typhus in mainland China: spatiotemporal expansion and risk prediction underpinned by complex factors. Emerg Microbes Infect. 2019 Jun 24; 8 (1):909–19. https://doi.org/10.1080/22221751.2019.1631719 PMID: 31233387
- Ashby J, Moreno-Madriñán MJ, Yiannoutsos CT, Stanforth A. Niche Modeling of Dengue Fever Using Remotely Sensed Environmental Factors and Boosted Regression Trees. Remote Sens. 2017 Apr; 9 (4):328.
- Forna A, Dorigatti I, Nouvellet P, Donnelly CA. Spatiotemporal variability in case fatality ratios for the 2013–2016 Ebola epidemic in West Africa. Int J Infect Dis IJID Off Publ Int Soc Infect Dis. 2020 Apr; 93:48–55. https://doi.org/10.1016/j.ijid.2020.01.046 PMID: 32004692
- Elith J, Leathwick JR, Hastie T. A working guide to boosted regression trees. J Anim Ecol. 2008 juillet; 77(4):802–13. https://doi.org/10.1111/j.1365-2656.2008.01390.x PMID: 18397250
- 26. Breiman L. Random Forests. Mach Learn. 2001 Oct 1; 45(1):5-32.
- 27. Meyer H, Reudenbach C, Hengl T, Katurji M, Nauss T. Improving performance of spatio-temporal machine learning models using forward feature selection and target-oriented validation. Environ Model Softw. 2018 Mar 1; 101:1–9.
- Pastore M. Overlapping: a R package for Estimating Overlapping in Empirical Distributions. J Open Source Softw. 2018 Dec 5; 3(32):1023.
- Pastore M, Calcagnì A. Measuring Distribution Similarities Between Samples: A Distribution-Free Overlapping Index. Front Psychol. 2019 May 21; 10:1089. <u>https://doi.org/10.3389/fpsyg.2019.01089</u> PMID: 31231264
- Rousseeuw PJ, Croux C. Alternatives to the Median Absolute Deviation. J Am Stat Assoc. 1993 Dec 1; 88(424):1273–83.
- Bivand RS, Wong DWS. Comparing implementations of global and local indicators of spatial association. TEST. 2018 Sep; 27(3):716–48.
- Durbin J, Watson GS. Testing Fot Serial Correlation In Least Squares Regression. II. Biometrika. 1951 Jun 1; 38(1–2):159–78. PMID: 14848121
- **33.** Lang M, Binder M, Richter J, Schratz P, Pfisterer F, Coors S, et al. mlr3: A modern object-oriented machine learning framework in R. J Open Source Softw. 2019 Dec 11; 4(44):1903.

- **34.** R Core Team. R: A language and environment for statistical computing [Internet]. Vienna, Austria: R Foundation for Statistical Computing; 2020. Available from: https://www.R-project.org/
- 35. Behrend MR, Basáñez MG, Hamley JID, Porco TC, Stolk WA, Walker M, et al. Modelling for policy: The five principles of the Neglected Tropical Diseases Modelling Consortium. PLoS Negl Trop Dis. 2020 Apr 9; 14(4):e0008033. https://doi.org/10.1371/journal.pntd.0008033 PMID: 32271755
- Stanaway JD, Roth G. The burden of Chagas disease: estimates and challenges. Glob Heart. 2015 Sep; 10(3):139–44. https://doi.org/10.1016/j.gheart.2015.06.001 PMID: 26407508
- Pinsent A, Solomon AW, Bailey RL, Bid R, Cama A, Dean D, et al. The utility of serology for elimination surveillance of trachoma. Nat Commun. 2018 Dec 21; 9(1):5444. <u>https://doi.org/10.1038/s41467-018-07852-0</u> PMID: 30575720
- Golden A, Faulx D, Kalnoky M, Stevens E, Yokobe L, Peck R, et al. Analysis of age-dependent trends in Ov16 IgG4 seroprevalence to onchocerciasis. Parasit Vectors. 2016; 9(1):338. https://doi.org/10.1186/ s13071-016-1623-1 PMID: 27296630